

7710 Collins Avenue
PB25-0802

Narrative Responses to Reviewer Comments

1. PB – Planning Department Review

In addition to the criteria that is already provided in the LOI, list and respond to the review criteria in sections 5.2.11(b)(3)(A)(I-II) and 5.2.11(f)(7).

Response: Please refer to the Letter of Intent included in this submittal, updated accordingly.

2. PB – Planning Admin Review

Page (5) The Power of Attorney must specifically identify the board the applicant is appearing before (e.g., Planning Board, Board of Adjustment, etc.). General or non-specific authorization will not be accepted.

Response: A corrected Power of Attorney document is included in this submittal. Please refer to the Application included in this submittal.